Author's response to reviews

Title: Weighing the evidence: obesity, metabolic syndrome, and the risk of chronic kidney disease

Authors:

Linda Shavit (lshavit@szmc.org.il)
Ezra Gabbay (gabbayez@hotmail.com)
Itzchak Slotki (islotki@szmc.org.il)

Version: 4 Date: 2 August 2015

Author's response to reviews: see over
To the Editor of BMC nephrology,

We thank the Editor for the additional careful review of our paper. Since there were several concerns expressed by the Editor and reviewers, we have assigned a number to each point and addressed them in numerical order. In the revised manuscript changes are indicated in red font.

1. Title page - It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author

Names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author have been added to the title page.

2. Please update the Abstract in the submission system.

Done

3. Acknowledgements - We strongly encourage you to include an Acknowledgements section between the Authors’ contributions section and Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements

We don’t have acknowledgements

b) Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

We don’t have funding for this study

4. Abbreviations - If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations can be provided, which should precede the competing interests and authors’ contributions.

Done